Author's response to reviews

Title: The incidence, aetiology and outcome of acute seizures in children admitted to a rural Kenyan district hospital

Authors:

Richard Idro (ridro@kilifi.kemri-wellcome.org)
Samson Gwer (sgwer@kilifi.kemri-wellcome.org)
Hellen Gatakaa (hgatakaa@kilifi.kemri-wellcome.org)
Moses Ndiritu (mndiritu@kilifi.kemri-wellcome.org)
Michael Kahindi (mkahindi@kilifi.kemri-wellcome.org)
Tony Kazungu (tkazungu@kilifi.kemri-wellcome.org)
Kathryn Maitland (kmaitland@kilifi.kemri-wellcome.org)
Brian GR Neville (b.neville@ich.ucl.ac.uk)
Piet A Kager (p.a.kager@amc.uvl.nl)
Charles RJC Newton (cnewton@kilifi.kemri-wellcome.org)

Version: 3 Date: 6 February 2008

Author's response to reviews: see over
5th February 2008

Dear Editor BMC Pediatrics,

Thank you for accepting our manuscript.

The following changes have been made as requested.

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.

These have been made.

Footnotes - We are unable to accommodate footnotes in the full text version of the manuscript. We suggest that the footnotes are incorporated into the text of the manuscript.

The footnotes have been removed.

Email address- please include email addresses for all the authors on the title page. Please use the following format-
GSF: g.s.feder@qmul.ac.uk
HS: hsohal@yahoo.co.uk
SE: s.eldridge@qmul.ac.uk

This change has been effected.

Section headings: Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT

Corrected

Author’s contributions- please change this to Authors’ contributions

This has been changed.

Competing interests - If there are none to declare, please write `The authors declare that they have no competing interests. Please move this section to above the authors' contributions

Declaration made.

Table orientation- we prefer tables and figures to fit onto a portrait A4 page. If possible please re-orientate your table to the portrait orientation.

The table tables have been re-oriented as suggested.
Figure title - The image file should not include the title (e.g. Figure 1... etc.) or figure number. The legend and title should be part of the manuscript file after the reference list. The figures are numbered automatically in the order in which they are uploaded.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

These changes have been made.

Please remove the addition files for the reviewers from the submission system.

These have been removed.

Tables- all tables need to be cited in the text in order (1 to 6). We could not find table 6 cited within the text. Please ensure you update the text to include a reference to table 6.

Table 5 was referenced twice. The second should have been table 6. I have now corrected this.

In addition, the minor changes requested by the reviewer about viridians Streptococci have been made.

Thank you

Richard Idro